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Stability run 209

• No major problems during the shift

• Observed a slight reduction of current for some RPC chambers in the new stability run (210), started yesterday evening (shown 

in the next slides)

2

 Humidity quite stable during the week

Stability run overview



Check list
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HV scan 729

Wednesday 3rd April

Data logger crashed when we started this run (just this time during the whole shift)

HV scans 729 and 730 very similar each other
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HV scan 730

Wednesday 3rd April

HV scans 729 and 730 very similar each other
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HV scan 728 from last week (for comparison)

No huge differences observed
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Ongoing stability run 210

• Started yesterday evening around 8 pm
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Currents in the new stability run (210)

• Observed a reduction of current for some detectors (RE1-1 TN, SHIP, KODEL-H-TOP, KODEL-H-BOT)

• HV, source attenuation filter, gas flow, gas mixture not changed

• Maybe due to a new setup in front of the source?



9

GIF++ setups on 28th March
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GIF++ setups on 4th April
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i0 settings

• We used the i0 settings taken from the checklist shared file


• But they are different w.r.t. the last stability run 209


• Which ones are correct? Can we keep updated the settings in the shared file?


